Author's response to reviews

Title: Macrophage traits in cancer cells are induced by macrophage-cancer cell fusion and cannot be explained by cellular interaction

Authors:

Ivan Shabo (ivan.shabo@liu.se)
Kristine Midtbø (krimi000@student.liu.se)
Henrik Andersson (Henrik.A.Andersson@lio.se)
Emma Åkerlund (emma.akerlund92@gmail.com)
Hans Olsson (hans.olsson@lio.se)
Pia Wegman (pia.wegman@lio.se)
Cecilia Gunnarsson (cecilia.gunnarsson@lio.se)
Annelie Lindström (annelie.lindstrom@liu.se)

Version: 6 Date: 14 October 2015

Author's response to reviews: see over
To the Editor of BMC Cancer

Dear Sir,

Thanks for good comments of our article entitled "Macrophage traits in cancer cells are induced by macrophage-cancer cell fusion and can not be explained by cellular interaction". The manuscript is revised as follows:

- **Minor essential revisions**: The figure references in page 8 (manuscript) is corrected according to recommendations from reviewer.

- **Discretionary revisions**: The figure references and text in page 8 (manuscript) are revised as recommended by the reviewer.

- E-mail addresses and affiliation of the authors are updated (page 1).

All of the above changes in the text were made in red text for easy tracking for you.

With best regards

Dr. Ivan Shabo, MD PhD

Consultant in endocrine and sarcoma surgery
Division of endocrine and sarcoma surgery
Department of Molecular Medicine and Surgery (MMK), K1 Karolinska Institution
Karolinska University Hospital, P9: 03
171 76 Stockholm
Sweden

Phone: +46 737772661
Mail: Ivan.Shabo@liu.se